Author's response to reviews

Title: Corticosterone mediates electroacupuncture-produced anti-edema in a rat model of inflammation

Authors:
- Aihui Li (ali002@umaryland.edu)
- Rui-Xin Zhang (rzhan001@umaryland.edu)
- Yi Wang (wangyi01101971@yahoo.com.cn)
- Haiqing Zhang (linneanzhang@hotmail.com)
- Ke Ren (kren@umaryland.edu)
- Brian M. Berman (bberman@compmed.umm.edu)
- Ming Tan (MTan@umm.edu)
- Lixing Lao (llao@compmed.umm.edu)

Version: 5 Date: 27 June 2007

Author's response to reviews: see over
Dear Dr. Parkin:

Attached please find the revised manuscript MS # 1685510606135173, entitled "Corticosterone mediates electroacupuncture-produced anti-edema in a rat model of inflammation" by Li et al. We have revised the manuscript carefully according to the editor’s suggestion. The following responses address the point-by-point changes made in the text.

1. Add Figure legends header for each figure legend (see page 15).

2. Change the symbols in the figure legend 1 into descriptions (see page 15, line 3-6).

Thank you!

Sincerely yours,

Lixing Lao, Ph.D., L.Ac.

Corresponding author
Center for Integrative Medicine
University of Maryland, School of Medicine
Kernan mansion, 3rd floor
2200 Kernan Drive
Baltimore, MD 21207-6697
Telephone: 410-448-6873
Fax: 410-448-6875
E-mail: llao@compmed.umm.edu